Author's response to reviews

Title: Copenhagen study of overweight patients with coronary artery disease undergoing low energy diet or interval training: The randomized CUT-IT trial protocol

Authors:

- Lene R Pedersen (lrpedersen@gmail.com)
- Rasmus H Olsen (rasmus.huan.olsen@regionh.dk)
- Marianne Frederiksen (marianne@hjertecafeen.dk)
- Arne Astrup (ast@life.ku.dk)
- Elizaveta Chabanova (Elizaveta.Chabanova@regionh.dk)
- Philip Hasbak (Philip.Hasbak@regionh.dk)
- Jens J Holst (holst@mfi.ku.dk)
- Andreas Kjær (akjaer@sund.ku.dk)
- John W Newman (John.Newman@ars.usda.gov)
- Rosemary Walzem (rwalzem@poultry.tamu.edu)
- Ulrik Wisløff (ulrik.wisloff@ntnu.no)
- Ahmad Sajadieh (Ahmad.Sajadieh@regionh.dk)
- Steen Haugaard (sbhau@dadlnet.dk)
- Eva Prescott (Eva.Irene.Bossano.Prescott@regionh.dk)

Version: 2
Date: 30 October 2013

Author's response to reviews: see over
Dear Editor

Thank you for considering our manuscript “Copenhagen study of overweight patients with coronary artery disease undergoing low energy diet or interval training: The randomised “CUT-IT” trial protocol” to BMC cardiovascular disorders.

I have made the requested changes:

- Provided the name of the ethics committee on page 13 in the Ethics and Dissemination section.

Furthermore, the requested documentation for approval by the Regional Ethics Committee of the Capital Region in Denmark and external funding from The Danish Council for Independent Research (The Danish Agency for Science, Technology and Innovation) has been sent to editorial@biomedcentral.com.

I am looking forward to hearing from you.

With kind regards

Lene Roerholm Pedersen, MD
Corresponding author